Identification of Atezolizumab plus BEvacizumab prognostic index via recursive partitioning

analysis in advanced hepatocellular carcinoma: the ABE index

Mara Persano^{1*}, Margherita Rimini^{2*}, Toshifumi Tada³, Goki Suda⁴, Shigeo Shimose⁵, Masatoshi Kudo⁶, Jaekyung Cheon⁷, Fabian Finkelmeier⁸, Ho Yeong Lim⁹, José Presa¹⁰, Gianluca Masi¹¹⁻¹², Changhoon Yoo¹³, Sara Lonardi¹⁴, Lorenza Rimassa¹⁵⁻¹⁶, Fabio Piscaglia¹⁷, Takashi Kumada¹⁸, Naoya Sakamoto⁴, Hideki Iwamoto⁵, Tomoko Aoki⁶, Hong Jae Chon⁷, Vera Himmelsbach⁸, Margarida Montes¹⁰, Caterina Vivaldi¹¹⁻¹², Caterina Soldà¹⁴, Atsushi Hiraoka¹⁹, Takuya Sho⁴, Takashi Niizeki⁵, Naoshi Nishida⁶, Christoph Steup⁸, Masashi Hirooka²⁰, Kazuya Kariyama²¹, Joji Tani²², Masanori Atsukawa²³, Koichi Takaguchi²⁴, Ei Itobayashi²⁵, Shinya Fukunishi²⁶, Kunihiko Tsuji²⁷, Toru Ishikawa²⁸, Kazuto Tajiri²⁹, Hironori Ochi³⁰, Satoshi Yasuda³¹, Hidenori Toyoda³¹, Chikara Ogawa³², Takashi Nishimura³³, Takeshi Hatanaka³⁴, Satoru Kakizaki³⁵, Noritomo Shimada³⁶, Kazuhito Kawata³⁷, Fujimasa Tada¹⁹, Hideko Ohama¹⁹, Kazuhiro Nouso²¹, Asahiro Morishita²², Akemi Tsutsui²⁴, Takuya Nagano²⁴, Norio Itokawa²³, Tomomi Okubo²³, Taeang Arai²³, Michitaka Imai²⁸, Hisashi Kosaka³⁸, Atsushi Naganuma³⁹, Yohei Koizumi²¹, Shinichiro Nakamura³, Masaki Kaibori³⁸, Hiroko Iijima³³, Yoichi Hiasa²⁰, Valentina Burgio², Angelo Della Corte⁴⁰,

Francesca Ratti⁴¹, Francesco De Cobelli⁴⁰, Luca Aldrighetti⁴¹, Mario Scartozzi¹, Stefano Cascinu⁴², Andrea Casadei-Gardini⁴²

¹ Medical Oncology, University and University Hospital of Cagliari, Cagliari, Italy.

² Department of Medical Oncology, IRCCS San Raffaele Hospital, Milan, Italy

³ Department of Internal Medicine, Japanese Red Cross Himeji Hospital, Himeji, Japan

⁴ Department of Gastroenterology and Hepatology, Graduate School of Medicine, Hokkaido University;

North 15, West 7, Kita-ku, Sapporo, Hokkaido 060-8638, Japan

⁵ Division of Gastroenterology, Department of Medicine, Kurume University School of Medicine, Kurume, Fukuoka 830-0011, Japan

⁶ Department of Gastroenterology and Hepatology, Kindai University Faculty of Medicine, Higashi-osaka

⁷ Department of Medical Oncology, CHA Bundang Medical Center, CHA University School of Medicine, Seongnam, Republic of Korea

⁸ Department of Internal Medicine 1, University Hospital Frankfurt, Goethe University, Frankfurt am Main, Germany

⁹ Department of Medicine, Samsung Medical Center, School of Medicine, Sungkyunkwan University, Seoul, South Korea

¹⁰ Liver Unit-CHTMAD, Vila Real, Portugal

¹¹ Unit of Medical Oncology 2, University Hospital of Pisa, Pisa, Italy

Commentato [RLH1]: Would you please add Tiziana Pressiani? Affiliation Medical Oncology and Hematology Unit, Humanitas Cancer Center, IRCCS Humanitas Research Hospital, Via A. Manzoni 56, 20089 Rozzano, Milan, Italy

Please move my name towards the end

¹² Department of Translational Research and New Technologies in Medicine and Surgery, University of Pisa, Pisa, Italy

¹³ Department of Oncology, ASAN Medical Center, University of Ulsan College of Medicine, 88, Olympicro 43-gil, Songpa-gu, Seoul 05505, Korea

¹⁴Oncology Unit 1, Veneto Institute of Oncology IOV-IRCCS, Padua, Italy

¹⁵ Department of Biomedical Sciences, Humanitas University, Via Rita Levi Montalcini 4, 20072 Pieve Emanuele, Milan, Italy

¹⁶ Medical Oncology and Hematology Unit, Humanitas Cancer Center, IRCCS Humanitas Research Hospital, Via A. Manzoni 56, 20089 Rozzano, Milan, Italy

¹⁷ Division of Internal Medicine, Hepatobiliary and Immunoallergic diseases, University of Bologna, Bologna, Italy

¹⁸ Department of Nursing, Gifu Kyoritsu University, Ogaki, Japan

¹⁹ Gastroenterology Center, Ehime Prefectural Central Hospital, Matsuyama, Japan

²⁰ Department of Gastroenterology and Metabology, Ehime University Graduate School of Medicine, Ehime, Japan

²¹ Department of Gastroenterology, Okayama City Hospital, Okayama, Japan

²² Department of Gastroenterology and Hepatology, Kagawa University, Kagawa, Japan

²³ Division of Gastroenterology and Hepatology, Department of Internal Medicine, Nippon Medical School, Tokyo, Japan

²⁴ Department of Hepatology, Kagawa Prefectural Central Hospital, Takamatsu, Japan

²⁵ Department of Gastroenterology, Asahi General Hospital, Asahi, Japan

²⁶ Department of Gastroenterology, Osaka Medical and Pharmaceutical University, Osaka, Japan

²⁷ Center of Gastroenterology, Teine Keijinkai Hospital, Sapporo, Japan

²⁸ Department of Gastroenterology, Saiseikai Niigata Hospital, Niigata, Japan

²⁹ Department of Gastroenterology, Toyama University Hospital, Toyama, Japan

³⁰ Hepato-biliary Center, Japanese Red Cross Matsuyama Hospital, Matsuyama, Japan

³¹ Department of Gastroenterology and Hepatology, Ogaki Municipal Hospital, Ogaki, Japan

³² Department of Gastroenterology, Japanese Red Cross Takamatsu Hospital, Takamatsu, Japan

³³ Department of Internal medicine, Division of Gastroenterology and Hepatology, Hyogo Medical University, Nishinomiya, Japan

³⁴ Department of Gastroenterology, Gunma Saiseikai Maebashi Hospital, Maebashi, Japan

³⁵ Department of Clinical Research, National Hospital Organization Takasaki General Medical Center, Takasaki, Japan

³⁶ Division of Gastroenterology and Hepatology, Otakanomori Hospital, Kashiwa, Japan

³⁷ Department of Hepatology, Hamamatsu University School of Medicine, Hamamatsu, Japan

³⁸ Department of Surgery, Kansai Medical University, Osaka, Japan

³⁹ Department of Gastroenterology, National Hospital Organization Takasaki General Medical Center, Takasaki, Japan

⁴⁰ School of Medicine, Vita-Salute San Raffaele University, Milan, Italy

⁴¹ Hepatobiliary Surgery Division, Liver Center, Vita-Salute San Raffaele University, IRCCS San Raffaele Scientific Institute, Milan, 20132, Italy

⁴² Department of Oncology, Vita-Salute San Raffaele University, IRCCS San Raffaele Scientific Institute Hospital, Milan, Italy

*Co-first authors

Corresponding Author: Margherita Rimini; Department of Medical Oncology, IRCCS San Raffaele Hospital, Via Olgettina n. 60, Milan, Italy; Email margherita.rimini@gmail.com

Electronic word count: 2900

Number of figures and tables: 3 figures, 2 tables.

Abbreviations: HCC: hepatocellular carcinoma; OS: overall survival; HR: hazard ratio; CI:

confidence interval; PD-L1: programmed cell death ligand-1; PFS: progression-free survival; ALBI:

albumin-bilirubin; NLR: neutrophil-lymphocyte ratio; RPA: recursive partitioning analysis; AE:

adverse events; AFP: alpha-fetoprotein; AST: aspartate aminotransferase; PD: progressive disease.

Conflict of Interest:

A.C.G. has received grants and personal fees from MSD, Eisai, Bayer, and is an advisor for MSD, Eisai, Bayer, Bristol-Myers Squibb, AstraZeneca and GSK.

M.S. is an advisor for MSD, Eisai, MERCK, SERVIER, and AMGEN.

M.K. has received grants from Taiho Pharmaceuticals, Chugai Pharmaceuticals, Otsuka, Takeda, Sumitomo Dainippon-Sumitomo, Daiichi Sankyo, AbbVie, Astellas Pharma, and Bristol-Myers Squibb; has received grants and personal fees from MSD, Eisai, and Bayer, and is an adviser for MSD, Eisai, Bayer, Bristol-Myers Squibb, Eli Lilly and ONO Pharmaceutical.

L.R. has received consulting fees from Amgen, ArQule, AstraZeneca, Basilea, Bayer, BMS, Celgene, Eisai, Exelixis, Genenta, Hengrui, Incyte, Ipsen, IQVIA, Lilly, MSD, Nerviano Medical Sciences, Roche, Sanofi, Servier, Taiho Oncology, Zymeworks; lecture fees from AbbVie, Amgen, Bayer, Eisai, Gilead, Incyte, Ipsen, Lilly, Merck Serono, Roche, Sanofi, <u>Servier</u>; travel expenses from AstraZeneca; and institutional research funding from Agios, ARMO BioSciences, AstraZeneca, BeiGene, Eisai, Exelixis, Fibrogen, Incyte, Ipsen, Lilly, MSD, Nerviano Medical Sciences, Roche, Zymeworks.

F.P. has received consulting or lecture fees from Consulting or lecture fees in the last two years from: Astrazeneca, Bayer, Bracco, EISAI, ESAOTE, Exact Sciences, IPSEN; MSD; Roche, Samsung, Tiziana Life Sciences.

The other authors declare no conflicts of interest.

Funding: The present work received no financial support.

Authors' contributions:

Conception and design: A. Casadei-Gardini, M. Persano, M. Rimini.

Acquisition of data (acquired and managed patients): All authors.

Analysis and interpretation of data: A. Casadei-Gardini, M. Persano, M. Rimini.

Writing, review, and/or revision of the manuscript: A. Casadei-Gardini, M. Persano, M. Rimini.

Final approval of manuscript: All authors.

Statement of Ethics

The study protocol was reviewed and approved by the local Ethics Committee (EC). In particular, the protocol was firstly approved by the San Raffaele Authority Hospital EC for the coordinating centre and subsequently approved by the remaining EC (number DSAN854-A-OS/5). A written

Commentato [RLH2]: If you may add Tiziana Pressiani please ask her for her COIs

informed consent was obtained according to the EC's recommendations. The study was conducted according to the ethical guidelines of the 1975 Declaration of Helsinki.

Acknowledgments: nothing to declare.

ABSTRACT

BACKGROUND & AIMS: this study aims to identify a new prognostic index by applying recursive partitioning analysis (RPA) in hepatocellular carcinoma (HCC) patients treated with atezolizumab plus bevacizumab (AB).

METHODS: RPA was applied on 784 consecutive HCC patients treated with AB.

RESULTS: RPA allowed the identification of the Atezolizumab BEvacizumab prognostic (ABE) index, comprising three groups of patients: low risk, [(i) Child-Pugh A (CPA) patients without macrovascular invasion (MVI) but with Albumin-Bilirubin (ALBI) 1, aspartate aminotransferase (AST) normal value (NV), and alpha-fetoprotein (AFP) < 400 ng/mL, (ii) CPA patients without MVI but with ALBI 1, AST increased value (IV), and neutrophil-lymphocyte ratio (NLR) < 3, and (iii) CPA patients with MVI, ALBI 1, and AFP < 400 ng/mL]; intermediate risk, [(i) CPA patients without MVI but with ALBI 1, AST NV, and AFP \geq 400 ng/mL, and (ii) CPA patients without MVI but with ALBI 1, AST IV, and NLR \geq 3]; high risk [(i) CPA patients with ALBI 2, (ii) CPA patients with ALBI 1, MVI, and AFP \geq 400 ng/mL, and (iii) CPA patients without MVI but with ALBI 1, MVI, and AFP \geq 400 ng/mL, and (iii) CPA patients with (CPA) patients with ALBI 1, MVI, and AFP \geq 400 ng/mL, and (iii) CPB patients]. Overall survival was 22.5 months [95% confidence interval (CI) 17.0-22.5 months] in patients with low risk (60.1%), 14.2 months (95% CI 12.4-15.7 months) in intermediate risk (19.1%), and 7.0 months (95% CI 6.0-8.7 months) in high risk (20.8%); low risk hazard ratio (HR) 1, intermediate risk HR 1.76 (95% CI 1.26-2.46), high risk HR 3.99 (95% CI 2.76-5.77); P < 0.01.

CONCLUSIONS: the ABE index is an easy-to-use tool able to stratify HCC patients undergoing first-line therapy with AB.

Commentato [RLH3]: Why "but"? Same comment for other "but" Commentato [RLH4]: It should be "within normal ranges"

Commentato [RLH5]: What is the advantage of this tool compared to consider just the high-risk features as in the IMbrave analysis? (from a reviewer point of you, not a criticism)

KEYWORDS: advanced hepatocellular carcinoma, atezolizumab plus bevacizumab, prognostic index, recursive partitioning analysis, ABE index

Lay summary: recursive partitioning analysis allowed the identification of the Atezolizumab BEvacizumab prognostic (ABE) index, comprising three groups of hepatocellular carcinoma patients: low risk, intermediate risk, and high risk. It is a useful tool because it can stratify patients who are candidates for atezolizumab plus bevacizumab from a prognostic point of view, identifying in an early manner those patients who could potentially benefit less from this therapeutic option.

INTRODUCTION

Hepatocellular carcinoma (HCC) remains the third leading cause of cancer death worldwide [1]. However, the therapeutic armamentarium available has expanded in recent years with the approval of new systemic treatments. Sorafenib was the only available first-line therapy for 10 years based on the results of the two trials, SHARP and Asia Pacific. In these two studies, overall survival (OS) was between 6.5 and 10.7 months, significantly longer than that obtained with placebo [2, 3]. Lenvatinib was the second drug approved in 2017 for the first-line treatment of HCC. The openlabel, multicenter, phase III REFLECT trial demonstrated non-inferiority of lenvatinib in OS (13.6 months) compared to sorafenib (12.3 months; hazard ratio [HR] 0.92; 95% confidence interval [CI] 0.79-1.06) [4]. The combination of the anti-programmed cell death ligand-1 (PD-L1) atezolizumab plus the anti-vascular-endothelial growth factor bevacizumab represents the first therapeutic doublet approved for HCC treatment in the first-line setting. Indeed, the IMbrave150 trial demonstrated that this combination can obtain an advantage in OS (19.2 v.s 13.4 months; P < 0.001) and progressionfree survival (PFS) (6.9 vs. 4.3 months; P < 0.001) compared to sorafenib [5]. Recently, the final data from the phase III HIMALAYA trial were published. The immunotherapeutic combination of

Commentato [RLH6]: I would streamline this, well known to everyone

ha eliminato:

the anti-PD-L1 durvalumab plus <u>a single priming dose of</u> the anti-cytotoxic T-lymphocyte antigen 4 tremelimumab was shown to significantly improve OS compared to sorafenib (16.4 vs. 13.8

In light of these results, lenvatinib and atezolizumab plus bevacizumab represent the two options

months; P = 0.0035) in the first-line setting [6].

currently most used in first-line therapy in clinical practice. It has become crucial for clinicians to have prognostic tools available to correctly frame patients and be able to offer them the best treatment.

Literature data highlighted some prognostic factors for HCC patients treated with lenvatinib or atezolizumab plus bevacizumab, such as Child Pugh score, Albumin-Bilirubin (ALBI) grade, and neutrophil-lymphocyte ratio (NLR) [7-19]. As regards lenvatinib, relative dose intensity, prognostic nutritional index, body mass index, and muscle mass have also proved to be important prognostic factors [20-23]. Recently, a complex prognostic index derived from a recursive partitioning analysis (RPA), the lenvatinib prognostic (LEP) index, has been validated [24, 25]. Regarding atezolizumab plus bevacizumab, there are still few real-world studies that evaluate potential prognostic factors in patients treated with this therapeutic combination. This study aims to identify a new prognostic index by applying RPA in HCC patients treated with atezolizumab plus bevacizumab in first-line setting.

METHODS

Patients

The overall cohort of this multicentric study included 784 consecutive HCC patients from 4 countries (Italy, Germany, Japan, and the Republic of Korea) treated with atezolizumab plus bevacizumab between October 2018 and April 2022. The study population was treated with atezolizumab plus bevacizumab as first-line therapy for Barcelona Clinic Liver Cancer (BCLC) B or C HCC, deemed not eligible for first treatment or re-treatment with surgical or loco-regional

Commentato [RLH7]: You may want to mention also the camrelizumab-apatinb phase 3 trial and the tiszelizumab phase 3 trial

Commentato [RLH8]: This is not correct, if you mention lenvatinib you should mention also sorafenib as in all international guidelines

Commentato [RLH9]: If you mention lenvatinib you should mention also sorafenib, the two drugs have the same role in international guidelines, BCLC, EASL position paper

Commentato [RLH10]: There are analyses from IMbrave, according to high-risk features, Vp4, varices, ALBI. They should be mentioned, see below for the references therapies. Eligible patients had HCC diagnosis confirmed histologically or confirmed clinically in accordance with international guidelines. None of them received previous systemic therapy. Due to the intent to build a real-word dataset of patients treated with atezolizumab plus bevacizumab, no specific inclusion and exclusion criteria were applied, except for the absence of previous systemic therapies for HCC. Atezolizumab plus bevacizumab was administered as described in the IMbrave150 trial, and all patients received 1200 mg of atezolizumab plus 15mg/kg of body weight of bevacizumab intravenously every 3 weeks [5]. Treatment interruptions and/or dose reductions were allowed to manage adverse events (AEs). AEs were graded using the National Cancer Institute Common Terminology Criteria for Adverse Events (NCI-CTCAE) version 5.0 [26].

The present study was approved by the ethics committee at each center, complied with the provisions of the Good Clinical Practice guidelines and the Declaration of Helsinki and local laws, and fulfilled the Regulation (EU) 2016/679 of the European Parliament and of the Council of 27 April 2016 on the protection of natural persons with regard to the processing of personal data.

Statistical analysis

Clinical features and hematologic blood tests were collected at baseline (the day before the start of treatment).

We applied survival tree regression to identify risk groups in the overall cohort after dichotomizing each variable. Beginning by the overall cohort, we conducted univariate Cox proportional hazards regression for each predictor variable. The criteria to define groups at each level comprised the variable with the highest HR. We later dichotomized the cohort using the selected predictor variable and replicated the univariate Cox models among each group. We recursively replayed this process among each new group till no variable met the criteria for selection. After establishing initial groups or "nodes", HRs for each node were calculated relative to the lowest risk node. Kaplan Meier methods were applied to estimate survival for each node, and an overall log-rank test was calculated for the model.

Commentato [RLH11]: Needed?

MedCalc package (MedCalc® version 16.8.4) was used for statistical analysis.

RESULTS

The clinical and laboratory characteristics of the study population are shown in Table 1. The first node split by the Child-Pugh class states that the survival difference between Child-Pugh A versus Child-Pugh B patients is greater than the difference between any other patients' subset. The same process has been recursively implemented to the arising subpopulations, giving origin to the partitioning tree represented in Figure 1. Between the 725 Child-Pugh A patients, the most meaningful split was by ALBI grade, among patients with ALBI grade 1 (n = 692) and patients with ALBI grade 2 (n = 33). Between the 692 Child-Pugh A patients with ALBI grade 1, the most considerable split was among patients with macrovascular invasion (MVI) (n = 153) and patients without macrovascular invasion (n = 539). Between the 153 Child-Pugh A patients with ALBI 1 and macrovascular invasion, the closing split was among patients with alpha-fetoprotein (AFP) < 400 ng/mL (n = 82) and patients with AFP \geq 400 ng/mL (n = 71). Between the 539 Child-Pugh A patients with ALBI 1 without macrovascular invasion, the most meaningful split was among patients with aspartate aminotransferase (AST) normal value (n = 223) and patients with AST increased value (n = 316). Within the latter patients' subset, the closing split was among patients with NLR < 3 (n = 199) and patients with NLR \geq 3 (n = 117). Between the 223 Child-Pugh A patients with ALBI 1 and AST normal value without macrovascular invasion, the closing split was among patients with AFP < 400 ng/mL (n = 190) and patients with AFP \ge 400 ng/mL (n = 33). In accordance with the RPA tree, we have found three groups of patients with differing outcomes with regard to OS. The first group, named "low risk", comprises (i) Child-Pugh A patients without macrovascular invasion but with ALBI 1, AST normal value, and AFP < 400 ng/mL, (ii) Child-Pugh A patients without macrovascular invasion but with ALBI 1, AST increased value, and NLR <

3, and (iii) Child-Pugh A patients with macrovascular invasion, ALBI 1, and AFP < 400 ng/mL.

Commentato [RLH12]: Change the following into MVI as in the abstract; also in figures and tables

Commentato [RLH13]: Already mentioned in full?

The second group, named "intermediate risk", comprises (i) Child-Pugh A patients without macrovascular invasion but with ALBI 1, AST normal value, and AFP \geq 400 ng/mL, and (ii) Child-Pugh A patients without macrovascular invasion but with ALBI 1, AST increased value, and NLR \geq 3. Finally, the third group, named "high risk", comprises (i) Child-Pugh A patients with ALBI 2, (ii) Child-Pugh A patients with ALBI 1, macrovascular invasion, and AFP \geq 400 ng/mL, and (iii) Child-Pugh B patients. We have denominated this score the Atezolizumab BEvacizumab prognostic index "ABE index".

OS was 22.5 months (95% CI 17.0-22.5 months) in patients with low risk (n = 471, 60.1%), 14.2 months (95% CI 12.4-15.7 months) in intermediate risk (n = 150, 19.1%), and 7.0 months (95% CI 6.0-8.7 months) in high risk (n = 163, 20.8%); low risk HR 1 (reference group), intermediate risk HR 1.76 (95% CI 1.26-2.46), high risk HR 3.99 (95% CI 2.76-5.77); P < 0.01 (Figure 2A). Receiver operating characteristic (ROC) curve analysis displayed an area under the curve (AUC) of 0.72 (95% CI 0.68-0.76; P < 0.0001).

PFS was 9.4 months (95% CI 8.4-10.8 months) in patients with low risk, 6.1 months (95% CI 5.5-8.1 months) in intermediate risk, and 5.3 months (95% CI 3.7-5.8 months) in high risk; low risk HR 1 (reference group), intermediate risk HR 1.47 (95% CI 1.14-1.89), high risk HR 1.79 (95% CI 1.37-2.35); P < 0.01 (Figure 2B).

The three groups had a differing rate of progressive disease (PD) at the first computed tomography (CT) response assessment (low risk 16.1%; intermediate risk 24.0%; high risk 29.4%; P < 0.01) and of patients who <u>received</u> a second line therapy (low risk 54.7%; intermediate risk 52.7%; high risk 31.9%; P < 0.01) (Figure 3A). Median OS in subsequent anticancer treatments was 10.5 months (95% CI 8.6-15.4 months) in patients with low risk, 7.9 months (95% CI 4.6-9.4 months) in intermediate risk, and 3.1 months (95% CI 2.4-3.7months) in high risk; low risk HR 1.00 (reference group), intermediate risk HR 1.57 (95% CI 1.07-2.31), high risk HR 3.36 (95% CI 2.17-5.19); P < 0.01 (Figure 3B).

Commentato [RLH14]: Do you think that high-risk features may have an impact on response and not only on survival (OS and PFS)? ha eliminato: performed

Commentato [RLH15]: I would say for

In the three groups, differing profiles of toxicity have been highlighted, notably in terms of hypertension (low risk 27.4%; intermediate risk 22.7%; high risk 17.2%, P = 0.03), proteinuria (low risk 28.7%; intermediate risk 35.3%; high risk 22.7%, P < 0.05), and hypothyroidism (low risk 6.1%; intermediate risk 2.7%; high risk 1.8%; P = 0.03) (Table 2).

DISCUSSION

By applying RPA to a real-world population of HCC patients treated in first-line setting with atezolizumab plus bevacizumab, we have created an easy-to-use prognostic index, the ABE index. In our analysis, low risk patients achieved better survival outcomes than the phase III IMbrave150 trial in terms of both OS (22.5 vs. 19.2 months) and PFS (9.4 vs. 6.9 months) [5]. The low risk group, therefore, allows us to identify patients who could potentially benefit most from this therapeutic combination even in terms of low percentages of PD at the first re-evaluation. This is also in line with literature data suggesting that HCC patients benefit most from systemic treatments if they are <u>administered</u> in the early stages of the disease when liver function is still well preserved [27-31]. This benefit also translates into median OS in subsequent anticancer treatments which in our data was 10.5 months in low risk patients and 3.1 months in high risk patients. Patients who had better outcomes from first-line therapy with atezolizumab plus bevacizumab maintained this greater benefit also in subsequent lines, most likely because they are those who, thanks to disease control, also maintained good liver function that allowed them to access the benefits of the following therapies. Low risk patients are also those who have received the therapeutic combination longer, resulting in greater exposure to the mechanisms determining the AEs' onset. Indeed, our data show that hypertension (typically caused by bevacizumab), as well as hypothyroidism (typically associated with immunotherapy), were more frequent in low risk patients than in the other groups. This phenomenon was also reported in a Japanese real-world study in which Child-Pugh A patients

Commentato [RLH16]: Any explanation?

Commentato [RLH17]: This should be discussed more

ha eliminato:

ha eliminato: performed

Pugh B patients, most likely because the latter have <u>received</u> the therapy for a shorter time [32]. Of particular importance for clinical practice are the prognostic factors characterizing the high risk group because they allow the identification of patients who derive limited benefits from therapy with atezolizumab plus bevacizumab (OS 7.0 and PFS 5.3 months). These factors are represented by Child-Pugh B, ALBI 2, and AFP \geq 400 ng/mL in patients with macrovascular invasion.

treated with atezolizumab plus bevacizumab reported a higher percentage of proteinuria than Child-

There is numerous evidence that the most important prognostic factors in HCC patients are those representatives of liver function [33-37]. In our analysis, the Child-Pugh score was the first split node that allowed us to stratify two populations. This is in line with literature data on this particular subset of patients treated with atezolizumab plus bevacizumab. In particular, a real-world study by D'Alessio and colleagues on 216 patients reported significant differences between patients with Child-Pugh A and patients with Child-Pugh B both in terms of OS (A 16.8 vs. B 6.7 months, P = (0.0003) and in terms of PFS (A 7.6 vs. B 3.4 months, P = (0.03) (38). Another real-world study conducted on 457 Japanese patients confirmed statistically significant differences in survival outcomes [OS: A not reached (NR) vs. B 6.4 months, P < 0.001; PFS: A 7.5 vs. B 6.0 months, P =0.011]. Tanaka et al also analyzed the differences between the individual scores constituting the Child-Pugh classes, highlighting progressively worse OS and PFS as the score increased [32]. This type of correlation was also found in patients receiving the other currently available first-line therapies, sorafenib and lenvatinib [7, 11, 33-35, 37]. The second split node within RPA was the ALBI score. Patients with Child-Pugh A but ALBI 2 composed the high risk group, having achieved an OS of 8.0 months. Tanaka and colleagues found that in patients with Child-Pugh A, OS and PFS worsen significantly from ALBI 1 to ALBI 2b [32]. Our work confirms that even in patients treated with atezolizumab plus bevacizumab, as with other available first-line therapies, Child-Pugh class and ALBI score represent the most important prognostic factors capable of identifying patients who are likely to have a limited benefit from first-line therapy and for which it

ha eliminato: performed

ha eliminato: very

Commentato [RLH18]: These are high-risk patients by definition

Commentato [RLH19]: But not in terms of response rate and safety, you have to discuss this

is necessary to carefully evaluate the relationship between possible benefits and risks associated with the treatment to be offered. Furthermore, the ABE index, being consistent with the literature data in real-world settings, is a reliable index to be used in clinical practice, even if further analyses on external cohorts of patients treated with atezolizumab plus bevacizumab are desirable for its validation.

Very interesting data is represented by the subsequent split node which helps to identify high risk patients. These are Child-Pugh A patients with ALBI 1 and macrovascular invasion. In these patients, the factor that caused the greatest difference in OS was the baseline AFP levels. Patients with AFP \geq 400 ng/mL composed the high risk group and achieved an OS of 7.6 months. In contrast, patients with low baseline AFP levels were classified as low risk having achieved an OS of 22.5 months. AFP is a known prognostic factor for HCC patients in both localized and advanced disease settings [7, 37, 39-42] and is part of another prognostic score, the CRAFITY score [43]. This score was obtained from the univariate and multivariate analyses performed on 190 HCC ha eliminato: z patients treated with immunotherapy. In multivariate analysis, macrovascular invasion was not a ha eliminato: the factor influencing the prognosis of these patients, while AFP \geq 100 ng/mL and C-reactive protein \geq 1 mg/dL were independent prognostic factors for OS also in multivariate analysis and constituted, in fact, the CRAFITY score. Patients who had neither of these two elevated values had OS of 27.6 months, while for patients with at least one of these criteria. OS was 11.3 and 6.4 months, ha eliminato: or both respectively. In contrast to our study, the population included in these analyses was highly ha eliminato: z heterogeneous because it included patients treated with immunotherapy in different lines and with ha eliminato: inhomogeneous different drugs. Furthermore, the cut-off used for AFP was not the one that has mostly been ha eliminato: always identified in the literature as a discriminant of the prognosis in patients with HCC [37]. ha eliminato: n A separate discussion deserves NLR, which represents the split node able to discriminate between

Child-Pugh A without macrovascular invasion but with ALBI 1 and increased AST values, those intermediate risk (NLR \geq 3: OS 13.9 months) from those low risk (NLR \leq 3: OS NR). NLR is a

ha eliminato: z

particularly interesting prognostic factor for patients being treated with immunotherapy because it is representative of the state of activation of the immune system [43-48]. Elevated NLR values can be determined by lymphopenia or neutrophilia. Lymphopenia can determine the ineffectiveness of the immune system's reaction to cancer cells [49]. Neutrophilia is associated with a high release of cytokines by macrophages recruited in the tumor microenvironment which contribute to the creation of a chronic inflammatory state, in which the perpetually active immune system undergoes functional exhaustion [50]. Regarding specifically HCC patients treated with atezolizumab plus bevacizumab, NLR was found to be an independent prognostic factor also in two Japanese real-world studies [18, 19].

Our study has some limitations, the most important being its retrospective nature so there are some gaps in the data collected. This limit is partly offset by the large sample size, the fact that the population is homogeneous from a therapeutic point of view, and that it is made up of both Western and Eastern patients. Another limitation is represented by the fact that it <u>is needed</u> to further validate this index in an external cohort of patients, and above all to investigate its predictive role, applying it to patients treated with other available first-line therapies, such as lenvatinib<u>and</u> sorafenib.

high-risk, Vp4, varices, and ALBI analyses from IMbrave, you cannot omit them in the discussion: -Finn RS et al. AACR 2021, abstr 5080, Cancer Res 2021;81(13_Suppl):Abstract nr CT009 -Breder VV et al. ASCO 2021, abstract 4073 (I have the poster if needed) -Merle P et al. APPLE 2021 (I have the slides if needed) -Kudo M et al. ILCA 2021, abstr O-18

Commentato [RLH20]: You have to add and discuss the

ha eliminato: could be useful

CONCLUSIONS

The "ABE index" is an easy-to-use tool because it is made up of factors widely used in clinical practice in HCC patients. It is also a useful tool because it can stratify HCC patients who are candidates for atezolizumab plus bevacizumab from a prognostic point of view, identifying in an early manner those patients who could potentially benefit less from this therapeutic option.

REFERENCES

Commentato [RLH21]: As above, which is the advantage of this tool compared to consider high-risk features per se?

- Llovet JM, Kelley RK, Villanueva A, et al. Hepatocellular carcinoma. Nat Rev Dis Primers 2021;7(1):6.
- Llovet JM, Ricci S, Mazzaferro V, et al. Sorafenib in advanced hepatocellular carcinoma. N Engl J Med 2008;359(4):378-390.
- Cheng AL, Kang YK, Chen Z, et al. Efficacy and safety of sorafenib in patients in the Asia-Pacific region with advanced hepatocellular carcinoma: a phase III randomised, doubleblind, placebo-controlled trial. Lancet Oncol 2009;10(1):25-34.
- Kudo M, Finn RS, Qin S, et al. Lenvatinib versus sorafenib in first-line treatment of patients with unresectable hepatocellular carcinoma: a randomised phase 3 non-inferiority trial. Lancet 2018;391(10126):1163-1173.
- Cheng AL, Qin S, Ikeda M, et al. Updated efficacy and safety data from IMbrave150: Atezolizumab plus bevacizumab vs. sorafenib for unresectable hepatocellular carcinoma. J Hepatol 2022;76(4):862-873.
- Ghassan K A-A, George L, Cheng A-L, et al. Tremelimumab plus durvalumab in unresectable hepatocellular carcinoma. NEJM 2022;1(8).
- Ueshima K, Nishida N, Hagiwara S, et al. Impact of baseline ALBI grade on the outcomes of hepatocellular carcinoma patients treated with lenvatinib: a multicenter study. Cancers 2019;11:952.
- Hatanaka T, Kakizaki S, Nagashima T, et al. Analyses of objective response rate, progression-free survival, and adverse events in hepatocellular carcinoma patients treated with lenvatinib: a multicenter retrospective study. Hepatol Res 2020;50:382–395.
- Sho T, Suda G, Ogawa K, et al. Lenvatinib in patients with unresectable hepatocellular carcinoma who do not meet the REFLECT trial eligibility criteria. Hepatol Res 2020;50:966–977.

Commentato [RLH22]: Add also the original publication in NEJM

Commentato [RLH23]: It is NEJM Evidence

- Ogushi K, Chuma M, Uojima H, et al. Safety and efficacy of lenvatinib treatment in Child-Pugh A and B patients with unresectable hepatocellular carcinoma in clinical practice: a multicenter analysis. Clin Exp Gastroenterol 2020;13:385–396.
- Maruta S, Ogasawara S, Ooka Y, et al. Potential of lenvatinib for an expanded indication from the REFLECT trial in patients with advanced hepatocellular carcinoma. Liver Cancer 2020;9:382–396.
- 12. Ohki T, Sato K, Kondo M, et al. Impact of adverse events on the progression-free survival of patients with advanced hepatocellular carcinoma treated with lenvatinib: a multicenter retrospective study. Drugs Real World Outcomes 2020;7:141–149.
- Hiraoka A, Kumada T, Atsukawa M, et al. Prognostic factor of lenvatinib for unresectable hepatocellular carcinoma in real-world conditions-multicenter analysis. Cancer Med 2019;8:3719–3728.
- Iwamoto H, Shimose S, Noda Y, et al. Initial experience of atezolizumab plus bevacizumab for unresectable hepatocellular carcinoma in real-world clinical practice. Cancers 2021;13:2786.
- Rimini M, Shimose S, Lonardi S, et al. Lenvatinib versus Sorafenib as first-line treatment in hepatocellular carcinoma: A multi-institutional matched case-control study. Hepatol Res 2021;51(12):1229-1241.
- Rapposelli IG, Tada T, Shimose S, et al. Adverse events as potential predictive factors of activity in patients with advanced hepatocellular carcinoma treated with lenvatinib. Liver Int 2021;41(12):2997-3008.
- Chon YE, Cheon J, Kim H, et al. Predictive biomarkers of survival in patients with advanced hepatocellular carcinoma receiving atezolizumab plus bevacizumab treatment. Cancer Med 2022;00:1-8.

- Eso Y, Takeda H, Taura K, Takai A, Takahashi K, Seno H. Pretreatment neutrophil-tolymphocyte ratio as a predictive marker of response to atezolizumab plus bevacizumab for hepatocellular carcinoma. Curr Oncol 2021;28:4157–4166.
- 19. Tada T, Kumada T, Hiraoka A, et al. Neutrophil-lymphocyte ratio predicts early outcomes in patients with unresectable hepatocellular carcinoma treated with atezolizumab plus bevacizumab: a multicenter analysis. Eur J Gastroenterol Hepatol 2022;34(6):698-706.
- Hiraoka A, Takashi K, Toshifumi T, et al. Nutritional index as prognostic indicator in patients receiving lenvatinib treatment for unresectable hepatocellular carcinoma. Oncology 2020;98(5):295-302.
- 21. Sasaki R, Fukushima M, Haraguchi M, et al. Response to lenvatinib is associated with optimal relative dose intensity in hepatocellular carcinoma: experience in clinical settings. Cancers 2019;11:1769.
- Uojima H, Chuma M, Tanaka Y, et al. Skeletal muscle mass influences tolerability and prognosis in hepatocellular carcinoma patients treated with lenvatinib. Liver Cancer 2020;9(2):193-206.
- 23. Hiraoka A, Kumada T, Kariyama K, et al. Clinical importance of muscle volume in lenvatinib treatment for hepatocellular carcinoma: analysis adjusted with inverse probability weighting. J Gastroenterol Hepatol 2021;36(7):1812-1819.
- Rimini M, Kang W, Burgio V, et al. Validation of the easy-to-use lenvatinib prognostic (LEP) index to predict prognosis in advanced hepatocellular carcinoma patients treated with lenvatinib. Hepatol Res 2022.
- 25. Rapposelli IG, Shimose S, Kumada T, et al. Identification of lenvatinib prognostic index via recursive partitioning analysis in advanced hepatocellular carcinoma. ESMO Open 2021;6(4):100190.

- 26. Freites-Martinez A, Santana N, Arias-Santiago S, Viera A. Using the Common Terminology Criteria for Adverse Events (CTCAE - Version 5.0) to Evaluate the Severity of Adverse Events of Anticancer Therapies. Actas Dermosifiliogr (Engl Ed). 2021;112(1):90-92.
- 27. Kudo M, Ueshima K, Chan S, et al. Lenvatinib as an initial treatment in patients with intermediate-stage hepatocellular carcinoma beyond up-to-seven criteria and child–pugh A liver function: a proof-of-concept study. Cancers (Basel) 2019;11(8):1084.
- 28. Shimose S, Kawaguchi T, Tanaka M, et al. Lenvatinib prolongs the progression-free survival time of patients with intermediate-stage hepatocellular carcinoma refractory to transarterial chemoembolization: A multicenter cohort study using data mining analysis. Oncol Lett 2020;20(3):2257-2265.
- 29. Casadei-Gardini A, Scartozzi M, Tada T, et al. Lenvatinib versus sorafenib in first-line treatment of unresectable hepatocellular carcinoma: an inverse probability of treatment weighting analysis. Liver Int 2021;00:1–9.
- 30. Shimose S, Iwamoto H, Tanaka M, et al. Alternating lenvatinib and trans-arterial therapy prolongs overall survival in patients with intermediate stage hepatocellular carcinoma: a propensity score matching study. Cancers 2021;13:160.
- Burgio V, Iavarone M, Di Costanzo GG, et al. Real-life clinical data of lenvatinib versus sorafenib for unresectable hepatocellular carcinoma in Italy. Cancer Manag Res 2021;13:9379-9389.
- 32. Tanaka T, Hiraoka A, Tada T, et al. Therapeutic efficacy of atezolizumab plus bevacizumab treatment for unresectable hepatocellular carcinoma in patients with Child-Pugh class A or B liver function in real-world clinical practice. Hepatol Res 2022;1–11.
- 33. Huynh J, Cho MT, Kim E J-H, et al. Post hoc analysis in patients (pts) with unresectable hepatocellular carcinoma (uHCC) who porgressed to Child- Pugh B (CPB) liver function in the phase III REFLECT studi of lenvatinib (LEN). J Clin Oncol 2021;39(3):298.

- 34. Terashima T, Yamashita T, Takata N, et al. Comparative analysis of liver functional reserve during lenvatinib and sorafenib for advanced hepatocellular carcinoma. Hepatol Res 2020;50(7):871-884.
- 35. Hiraoka A, Kumada T, Atsukawa M, et al. Early relative change in hepatic function with lenvatinib for unresectable hepatocellular carcinoma. Oncology 2019;97(6):334-340.
- 36. Briggs A, Daniele B, Dick K, et al. Covariate-adjusted analysis of the phase 3 REFLECT study of lenvatinib versus sorafenib in the treatment of unresectable hepatocellular carcinoma. Br J Cancer 2020;122:1754-1759.
- Marisi G, Cucchetti A, Ulivi P, et al. Ten years of sorafenib in hepatocelllar carcinoma: are there any predictive and/or prognostic markers? World J Gastroenerol 2018;24(36):4152-4163.
- 38. D'Alessio A, Fulgenzi CAM, Nishida N, et al. Preliminary evidence of safety and tolerability of atezolizumab plus bevacizumab in patients with hepatocellular carcinoma and Child-Pugh A and B cirrhosis: a real-world study. Hepatology 2022;00:1–13.
- Hiraoka A, Kumada T, Kariyama K, et al. Clinical features of lenvatinib for unresectable hepatocellular carcinoma in real-world conditions: multicenter analysis. Cancer Med 2019;8:137–146.
- 40. Kodama K, Kawaoka T, Namba M, et al. Correlation between early tumor marker response and imaging response in patients with advanced hepatocellular carcinoma treated with lenvatinib. Oncology 2019;97(2):75-81.
- Saeki I, Yamasaki T, Yamashita S, et al. Early predictors of objective response in patients with hepatocellular carcinoma undergoing lenvatinib treatment. Cancers (Basel) 2020;12(4):779.
- 42. Liu B, Shang X, Shi J-Y, et al. Early Alpha-fetoprotein response is associated with survival in patients with hbv-related hepatocellular carcinoma receiving lenvatinib. Front Oncol 2022;12:807189.

- 43. Absenger G, Szkandera J, Pichler M, et al. A derived neutrophil to lymphocyte ratio predicts clinical outcome in stage II and III colon cancer patients. Br J Cancer 2013;109:395–400.
- Balta S, Unlu M, Arslan Z, Demirkol S. Neutrophil-to-lymphocyte ratio in prognosis of gastric cancer. J Gastric Cancer 2013;13:196–197.
- 45. Dirican A, Kucukzeybek BB, Alacacioglu A, et al. Do the derived neutrophil to lymphocyte ratio and the neutrophil to lymphocyte ratio predict prognosis in breast cancer? Int J Clin Oncol 2015;20:70–81.
- 46. Feng JF, Huang Y, Liu JS. Combination of neutrophil lymphocyte ratio and platelet lymphocyte ratio is a useful predictor of postoperative survival in patients with esophageal squamous cell carcinoma. Onco Targets Ther 2013;6:1605–1612.
- Ohno Y, Nakashima J, Ohori M, Gondo T, Hatano T, Tachibana M. Followup of neutrophilto-lymphocyte ratio and recurrence of clear cell renal cell carcinoma. J Urol 2012;187:411-417.
- 48. Paramanathan A, Saxena A, Morris DL. A systematic review and meta-analysis on the impact of pre-operative neutrophil lymphocyte ratio on long term outcomes after curative intent resection of solid tumours. Surg Oncol 2014;23:31–39.
- Ohtani, H. Focus on TILs: prognostic significance of tumor infiltrating lymphocytes in human colorectal cancer. Cancer Immun 2007;7:4.
- Kuang DM, Zhao Q, Wu Y, et al. Peritumoral neutrophils link inflammatory response to disease progression by fostering angiogenesis in hepatocellular carcinoma. J Hepatol 2011;54:948–955.

Legends:

Figure 1. Classification of patients treated with atezolizumab plub bevacizumab according to RPA. RPA identifies three risk group: low risk (in green), [(i) Child-Pugh A pt without MV

invasion but with ALBI 1, AST NV, and AFP < 400 ng/mL, (ii) Child-Pugh A pt without MV invasion but with ALBI 1, AST IV, and NLR < 3, and (iii) Child-Pugh A pt with MV invasion, ALBI 1, and AFP < 400 ng/mL;]; intermediate risk (in blue), [(i) Child-Pugh A pt without MV invasion but with ALBI 1, AST NV, and AFP \geq 400 ng/mL, and (ii) Child-Pugh A pt without MV invasion but with ALBI 1, AST IV, and NLR \geq 3]; high risk (in orange) [(i) Child-Pugh A pt with ALBI 2, (ii) Child-Pugh pt with ALBI 1, MV invasion, and AFP \geq 400 ng/mL, and (iii) Child-Pugh B pt]. AFP: alpha-fetoprotein; ALBI: Albumin-Bilirubin; AST: aspartate aminotransferase; IV: increased value; mo: months; MV: macrovascular; NLR: neutrophillymphocyte ratio; NR: not reached; NV: normal value; OS: overall survival; pt: patients; RPA: recursive partitioning analysis.

Figure 2. Median OS (A) and PFS (B) of patients treated with atezolizumab plus bevacizumab based on risk groups identified with RPA. OS: overall survival; PFS: progression-free survival; RPA: recursive partitioning analysis.

Figure 3. DCR and median OS in subsequent anticancer treatments of patients treated with atezolizumab plus bevacizumab based on risk groups identified with RPA. DCR: disease control rate; OS: overall survival; PD: progressive disease; RPA: recursive partitioning analysis.

Table 1. Baseline characteristics of study population. AFP: alpha-feto-protein; ALBI: Albumin-Bilirubin; BCLC: Barcelona Clinic Liver Cancer; CP: Child Pugh; EHD: extrahepatic disease; MV: macrovascular; NLR: neutrophil-lymphocyte ratio; PS: performance status.

Table 2. Adverse events reported in patients treated with atezolizumab plus bevacizumab.